Author's response to reviews

Title: Effects of Protein-Enriched Meal in a Weight Loss Program on Liver, Kidney or Bone: a Randomized Controlled Trial

Authors:

Zhaoping Li (zli@mednet.ucla.edu)
Leo Treyzon (ltreyzon@hotmail.com)
Steve Chen (steve.chen@amylin.com)
Eric Yan (ericyanmd@yahoo.com)
Gail Thames (gthames@mednet.ucla.edu)
Catherine Carpenter (ccarpenter@mednet.ucla.edu)

Version: 2 Date: 2 July 2010

Author's response to reviews: see over
July 2, 2010

Editor-in-Chief
Nutrition Journal

Dear Editor-in-Chief,

My colleagues and I believe we can satisfactorily rebut the comments of the reviewers and provide a re-written manuscript with appropriate changes. We are resubmitting a revised manuscript for your review.

Sincerely yours,

Zhaoping Li, MD, PhD
Professor of Medicine
Center for Human Nutrition
David Geffen School of Medicine at UCLA
Box 951742
Los Angeles, CA 90095
Tel: (310) 206-1987
Fax: (310) 206-5264
zlr@mednet.ucla.edu